# Telmisartan to reduce insulin resistance in HIV-positive individuals on combination antiretroviral therapy: the TAILoR dose-ranging Phase II RCT

Sudeep Pushpakom,<sup>1†</sup> Ruwanthi Kolamunnage-Dona,<sup>2†</sup> Claire Taylor,<sup>3</sup> Terry Foster,<sup>1</sup> Catherine Spowart,<sup>3</sup> Marta Garcia-Finana,<sup>2</sup> Graham J Kemp,<sup>4</sup> Thomas Jaki,<sup>5</sup> Saye Khoo,<sup>1</sup> Paula Williamson<sup>2</sup> and Munir Pirmohamed<sup>1\*</sup> for the TAILoR Study Group<sup>‡</sup>

Declared competing interests of authors: Thomas Jaki has received grants from the NIHR (National Institute for Health Research) Health Technology Assessment programme and funding from the pharmaceutical industry [Roche (F.Hoffmann-La Roche Ltd, Basel, Switzerland), as a member of an Independent Data Monitoring Committee, a post-doctoral fellowship and as a member of the European Union (EU) project Improving Design, Evaluation and Analysis of early drug development Studies (IDEAS); Bayer (Bayer AG, Leverkusen, Germany) as a member of the EU project IDEAS; Novartis (Novartis International AG, Basel, Switzerland), as a member of the EU project IDEAS; Janssen (Janssen Pharmaceutica NV, Beerse, Belgium), for presentation at an internal conference and as a member of the EU project IDEAS; AstraZeneca (AstraZeneca Plc, Cambridge, UK), as a member of the EU project IDEAS; and Baxalta (Baxalta now part of Shire Plc, St Helier, Jersey), for a research project]. Saye Khoo has received grants from the pharmaceutical industry [Merck (Merck & Co. Inc., NJ, USA), Gilead (Gilead Sciences Inc., CA, USA), ViiV Healthcare (ViiV Healthcare, Middlesex, UK) and Janssen, as support for Liverpool human immunodeficiency virus Drug Interactions website; and Viiv and Merck, as support for research projects]. Paula Williamson has received income from AbbVie (AbbVie Inc., IL, USA) related to fees for the preparation of a clinical study report for the randomised controlled trial of the clinical effectiveness, safety and cost-effectiveness of adalimumab in combination with methotrexate for the treatment of juvenile idiopathic arthritis-associated uveitis (SYCAMORE). Munir Pirmohamed is programme director for the MRC Clinical Pharmacology Training Scheme which funded jointly by the Medical Research Council and industry [Roche, UCB (UCB, Brussels, Belgium), Novartis and Eli Lilly (Eli Lilly and Company, IN, USA)].

<sup>&</sup>lt;sup>1</sup>Department of Molecular and Clinical Pharmacology, University of Liverpool, Liverpool, UK

<sup>&</sup>lt;sup>2</sup>Department of Biostatistics, University of Liverpool, Liverpool, UK

<sup>&</sup>lt;sup>3</sup>Clinical Trials Research Centre, University of Liverpool, Liverpool, UK

<sup>&</sup>lt;sup>4</sup>Liverpool Magnetic Resonance Imaging Centre, University of Liverpool, Liverpool, UK

<sup>&</sup>lt;sup>5</sup>Department of Mathematics and Statistics, Lancaster University, Lancaster, UK

<sup>\*</sup>Corresponding author munirp@liverpool.ac.uk

<sup>†</sup>Joint authors

<sup>&</sup>lt;sup>‡</sup>For names of TAILoR Study Group, please refer to Appendix 1

Published July 2019 DOI: 10.3310/eme06060

# **Plain English summary**

### The TAILoR dose-ranging Phase II RCT

Efficacy and Mechanism Evaluation 2019; Vol. 6: No. 6 DOI: 10.3310/eme06060

NIHR Journals Library www.journalslibrary.nihr.ac.uk

## **Plain English summary**

uman immunodeficiency virus (HIV) infection in humans is now a chronic disease that is treatable by a combination of anti-HIV drugs. This has resulted in a reduction in HIV-related deaths, but it has also led to the emergence of serious side effects, such as HIV-associated lipodystrophy, diabetes mellitus and, importantly, an increase in the risk of ischaemic heart disease. A key abnormality seems to be insulin resistance. There is a need to find new strategies to reduce insulin resistance in individuals infected with HIV, which would ultimately reduce the associated cardiovascular risk.

In the current randomised clinical trial, we have investigated whether or not telmisartan, a drug that is widely used for hypertension, can reduce insulin resistance in individuals infected with HIV on anti-HIV drugs. We used a novel adaptive trial design to compare three different doses of telmisartan with the control group (those individuals who do not take telmisartan) in the initial stage. We then selected the best telmisartan dose for the second stage, in which it was tested against the control to determine the effect on insulin resistance. We also tested the effect of telmisartan on body, liver and limb fat, blood proteins (markers of metabolic disease) and urine proteins.

A 80-mg dose of telmisartan was taken forward into the second stage of the study. Telmisartan did not reduce the primary marker of insulin resistance [homeostatic model assessment of insulin resistance (HOMA-IR)] over 24 weeks. It also did not affect the levels of lipids or hormones produced by fat cells. However, over 48 weeks, it led to marginal improvements in another marker of insulin resistance [revised Quantitative Insulin Sensitivity Check Index (QUICKI)], a marker of inflammation (high-sensitivity C-reactive protein) and reduced protein excretion from kidneys. Magnetic resonance imaging analysis showed a reduction in liver fat content. Overall, we did not show an effect of telmisartan on our primary marker of insulin resistance in individuals infected with HIV who are on antiretroviral drugs.

## **Efficacy and Mechanism Evaluation**

ISSN 2050-4365 (Print)

ISSN 2050-4373 (Online)

This journal is a member of and subscribes to the principles of the Committee on Publication Ethics (COPE) (www.publicationethics.org/).

Editorial contact: journals.library@nihr.ac.uk

The full EME archive is freely available to view online at www.journalslibrary.nihr.ac.uk/eme. Print-on-demand copies can be purchased from the report pages of the NIHR Journals Library website: www.journalslibrary.nihr.ac.uk

#### Criteria for inclusion in the Efficacy and Mechanism Evaluation journal

Reports are published in *Efficacy and Mechanism Evaluation* (EME) if (1) they have resulted from work for the EME programme, and (2) they are of a sufficiently high scientific quality as assessed by the reviewers and editors.

#### **EME** programme

The Efficacy and Mechanism Evaluation (EME) programme was set up in 2008 as part of the National Institute for Health Research (NIHR) and the Medical Research Council (MRC) coordinated strategy for clinical trials. The EME programme is broadly aimed at supporting 'science driven' studies with an expectation of substantial health gain and aims to support excellent clinical science with an ultimate view to improving health or patient care.

Its remit includes evaluations of new treatments, including therapeutics (small molecule and biologic), psychological interventions, public health, diagnostics and medical devices. Treatments or interventions intended to prevent disease are also included.

The EME programme supports laboratory based or similar studies that are embedded within the main study if relevant to the remit of the EME programme. Studies that use validated surrogate markers as indicators of health outcome are also considered.

For more information about the EME programme please visit the website: http://www.nets.nihr.ac.uk/programmes/eme

#### This report

The research reported in this issue of the journal was funded by the EME programme as project number 10/60/37. The contractual start date was in April 2012. The final report began editorial review in June 2017 and was accepted for publication in December 2017. The authors have been wholly responsible for all data collection, analysis and interpretation, and for writing up their work. The EME editors and production house have tried to ensure the accuracy of the authors' report and would like to thank the reviewers for their constructive comments on the final report document. However, they do not accept liability for damages or losses arising from material published in this report.

This report presents independent research. The views and opinions expressed by authors in this publication are those of the authors and do not necessarily reflect those of the NHS, the NIHR, the MRC, NETSCC, the EME programme or the Department of Health and Social Care. If there are verbatim quotations included in this publication the views and opinions expressed by the interviewees are those of the interviewees and do not necessarily reflect those of the authors, those of the NHS, the NIHR, NETSCC, the EME programme or the Department of Health and Social Care.

© Queen's Printer and Controller of HMSO 2019. This work was produced by Pushpakom et al. under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This issue may be freely reproduced for the purposes of private research and study and extracts (or indeed, the full report) may be included in professional journals provided that suitable acknowledgement is made and the reproduction is not associated with any form of advertising. Applications for commercial reproduction should be addressed to: NIHR Journals Library, National Institute for Health Research, Evaluation, Trials and Studies Coordinating Centre, Alpha House, University of Southampton Science Park, Southampton SO16 7NS, UK.

Published by the NIHR Journals Library (www.journalslibrary.nihr.ac.uk), produced by Prepress Projects Ltd, Perth, Scotland (www.prepress-projects.co.uk).

#### **NIHR Journals Library Editor-in-Chief**

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

#### **NIHR Journals Library Editors**

**Professor John Powell** Chair of HTA and EME Editorial Board and Editor-in-Chief of HTA and EME journals. Consultant Clinical Adviser, National Institute for Health and Care Excellence (NICE), UK, and Honorary Professor, University of Manchester, and Senior Clinical Researcher and Associate Professor, Nuffield Department of Primary Care Health Sciences, University of Oxford, UK

**Professor Andrée Le May** Chair of NIHR Journals Library Editorial Group (HS&DR, PGfAR, PHR journals) and Editor-in-Chief of HS&DR, PGfAR, PHR journals

**Professor Matthias Beck** Professor of Management, Cork University Business School, Department of Management and Marketing, University College Cork, Ireland

Dr Tessa Crilly Director, Crystal Blue Consulting Ltd, UK

Dr Eugenia Cronin Senior Scientific Advisor, Wessex Institute, UK

Dr Peter Davidson Consultant Advisor, Wessex Institute, University of Southampton, UK

Ms Tara Lamont Director, NIHR Dissemination Centre, UK

**Dr Catriona McDaid** Senior Research Fellow, York Trials Unit, Department of Health Sciences, University of York, UK

Professor William McGuire Professor of Child Health, Hull York Medical School, University of York, UK

Professor Geoffrey Meads Professor of Wellbeing Research, University of Winchester, UK

Professor John Norrie Chair in Medical Statistics, University of Edinburgh, UK

**Professor James Raftery** Professor of Health Technology Assessment, Wessex Institute, Faculty of Medicine, University of Southampton, UK

Dr Rob Riemsma Reviews Manager, Kleijnen Systematic Reviews Ltd, UK

Professor Helen Roberts Professor of Child Health Research, UCL Great Ormond Street Institute of Child Health, UK

Professor Jonathan Ross Professor of Sexual Health and HIV, University Hospital Birmingham, UK

**Professor Helen Snooks** Professor of Health Services Research, Institute of Life Science, College of Medicine, Swansea University, UK

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

**Professor Jim Thornton** Professor of Obstetrics and Gynaecology, Faculty of Medicine and Health Sciences, University of Nottingham, UK

Professor Martin Underwood Warwick Clinical Trials Unit, Warwick Medical School, University of Warwick, UK

Please visit the website for a list of editors: www.journalslibrary.nihr.ac.uk/about/editors

Editorial contact: journals.library@nihr.ac.uk